Tropical rains controlling deposition of Saharan dust across the North Atlantic Ocean

Michèlle van der Does^{1*}, Geert-Jan A. Brummer^{1,2}, Fleur C.J. van Crimpen^{1,3}, Laura F.

Korte¹, Natalie M. Mahowald⁴, Ute Merkel⁵, Hongbin Yu⁶, Paquita Zuidema⁷, Jan-Berend W. Stuut^{1,2,5}

¹ NIOZ, Royal Netherlands Institute for Sea Research, Department of Ocean Systems, and Utrecht University, Texel, The Netherlands

- ² Department of Earth Sciences, Faculty of Science, Vrije Universiteit Amsterdam, Amsterdam, The Netherlands
 ³ Faculty of Science, University of Amsterdam, Amsterdam, The Netherlands
 - ⁴ Department of Earth and Atmospheric Sciences, Cornell University, Ithaca, NY, USA
 - ⁵ MARUM Center for Marine Environmental Sciences, University of Bremen, Bremen, Germany
- ⁶ Earth Sciences Division, NASA Goddard Space Flight Center, Greenbelt, MD, USA

⁷ Rosenstiel School of Marine and Atmospheric Science, University of Miami, Miami, FL, USA

* Now at Alfred Wegener Institute, Helmholtz Centre for Polar and Marine Research, Bremerhaven, Germany. Correspondence to: michelle.van.der.does@awi.de

20

5

10

15

Geophysical Research Letters 47, 2020. https://doi.org/10.1029/2019GL086867

Key Points

25 1. Dust deposition fluxes collected while settling in the tropical North Atlantic show highest dust deposition in summer and autumn

2. Wet deposition of Saharan dust dominates over the tropical North Atlantic from July to October

3. Model simulations corroborate sediment trap and satellite observations of wet and dry dust deposition

30

Keywords

Mineral dust Atlantic Ocean

Dust deposition

35 Wet deposition

Ocean fertilization

Abstract

- 40 Mineral dust plays an important role in the atmospheric radiation budget as well as in the ocean carbon cycle through fertilization and by ballasting of settling organic matter. However, observational records of open-ocean dust deposition are sparse. Here, we present the spatial and temporal evolution of Saharan dust deposition over 2 years from marine sediment traps across the North Atlantic, directly below the core of the Saharan dust plume, with highest dust fluxes observed in summer. We combined
- 45 the observed deposition fluxes with model simulations and satellite observations and argue that dust deposition in the Atlantic is predominantly controlled by summer rains. The dominant depositional pathway changes from wet deposition in summer to dry deposition in winter. Wet deposition has previously been suggested to increase the release of dust-derived nutrients and their bioavailability, which may be a key contributor to surface-ocean productivity in remote and oligotrophic parts of the
- 50 oceans.

Plain Language Summary

Large quantities of dust are emitted from the Sahara Desert and blown over the (sub)tropical North Atlantic Ocean. Atmospheric dust plays an important role in the Earth's climate, but observations of

- 55 open-ocean deposition are sparse. Here, we present a continuous, 2-year record of dust deposition across the North Atlantic Ocean, between Africa and the Caribbean. We compare and evaluate measurements of dust deposition with model simulations and satellite observations. The combined results reveal that tropical rains dominate dust deposition across the Atlantic. In particular, the rain is responsible for the high dust deposition during summer and may also affect the nutrients that are
- 60 carried with the dust. Since previous studies suggest that the nutrients that reach the surface ocean with the rain are more readily available for phytoplankton productivity, the wet deposition of dust potentially has a major impact on the ocean carbon cycle.

1. Introduction

- Large quantities of mineral dust are blown from the Sahara Desert across the Atlantic (Yu et al., 2015). This dust impacts the atmosphere by altering the atmospheric radiation budget (Ryder et al., 2013) and by serving as cloud condensation nuclei (Twohy et al., 2009) and ice nuclei (Atkinson et al., 2013). Dust deposition also influences the ocean carbon cycle through the delivery of nutrients, which stimulate phytoplankton growth (Mills et al., 2004; Pabortsava et al., 2017) and by ballasting organic
 particles settling to the ocean floor (Bressac et al., 2014; Pabortsava et al., 2017; Van der Jagt et al., 2018). These effects on the atmosphere and ocean depend on atmospheric dust concentrations and deposition fluxes, and the size, shape and mineralogy of the dust particles (Shao et al., 2011). Dust particle size varies seasonally along the transect (Van der Does et al., 2016), which could in turn be related to compositional variations along the downwind trajectory as a result of preferential settling of
- coarser and heavier dust particles, related to their mineralogy (Korte et al., 2017).

The seasonal variability of dust transport and deposition over the Atlantic is subject to many factors. According to remote-sensing observations, maximum dust emission occurs in summer, while dust emissions are at a minimum in winter and autumn (Adams et al., 2012). Several dust-transporting

- 80 wind systems are active, transporting the dust at different altitudes in the atmosphere during the different seasons (Stuut et al., 2005; Muhs, 2013; Yu et al., 2019). First, dust transport occurs year-round in the shallow north-easterly trade-wind layer. Furthermore, dust transport is influenced by the intertropical convergence zone (ITCZ; Nicholson, 2000), with the center of the main dust cloud migrating from 10–20°N in summer to 0–10°N in winter (Adams et al., 2012). With the southward
- 85 migration of the ITCZ in winter the Harmattan winds intensify, transporting dust at altitudes between 0 and 3 km over the Atlantic (Muhs, 2013; Tsamalis et al., 2013). During summer when the ITCZ

migrates north, dust transport across the Atlantic occurs within the Saharan Air Layer at maximum altitudes of 5 to 7 km (Muhs, 2013; Weinzierl et al., 2017). The particle size of the deposited dust shows coarser grains at proximal locations, and finer-grained material at more distant sites (Mahowald

90 et al., 2014; Van der Does et al., 2016), although giant mineral dust particles (>75 μm) were also identified thousands of kilometres from their sources (Van der Does et al., 2018).

The magnitude of dust deposition over the ocean is a function of the travelled distance, with more deposition close to the source (Sarnthein et al., 1981; Goudie & Middleton, 2001). Atmospheric dust particles can be removed from the atmosphere by dry and wet depositional processes (Zender et al., 2003; Bergametti & Forêt, 2014). Atmospheric processing of dust is of great interest since dust

- processing in aerosols under low pH conditions favors the leaching of important nutrients (Meskhidze et al., 2005; Korte et al., 2018), while the iron in dust is made more soluble in clouds due to reactions with oxalate (Johnson & Meskhidze, 2013), which are subsequently released into the surface ocean
- 100 (Spokes et al., 1994) and lead to increased primary production and chlorophyll-a (Ridame et al., 2014). This may lead to higher solubility of iron and other nutrients in wet deposition (Spokes et al., 1994). In contrast, a study by Korte et al. (2018) suggests that dry-deposited dust has a much lower nutrient bio-availability. Also, since satellite instruments cannot detect wet dust deposition below the clouds, a multidisciplinary approach combining *in situ* observations and satellite remote sensing with model
 105 simulations should provide a better understanding of processes involved in dust transport and
 - deposition type, and help to constrain the biogeochemical impact of dust deposition.

So far, transport and deposition of Saharan dust across the Atlantic Ocean was studied at sea during research expeditions (Stuut et al., 2005; Baker et al., 2010; Weinzierl et al., 2017), or monitored by
sediment-trap collection at stations only at the eastern (Bory & Newton, 2000; Neuer et al., 2004; Brust et al., 2011) and western sides of the Atlantic (Carlson & Prospero, 1972; Jickells et al., 1998), and estimated by remote sensing (Yu et al., 2015). Here, we present Saharan dust deposition fluxes collected while settling in the tropical North Atlantic Ocean, along a transect below the core of the Saharan dust plume (at 12°N).

115

120

95

2. Methods

We monitored 2 years of Saharan dust deposition fluxes at four sites across the Atlantic Ocean using time-series from submarine sediment traps (M1–M4). The traps were moored at ~1,200 m water depth, three positioned along a transect at 12°N, and a fourth at 13°N (map Fig. 3), which sampled synchronously from October 2012 to October 2014 (Table S1). The transect is complemented by 2

- years (2013–2014) of Saharan dust concentrations measured on Barbados (13°N, 59°W). We combined these observational data with dry- and wet-dust deposition fluxes simulated using the Community Earth System Model (CESM 1.2.2; Hurrell et al., 2013), and dust optical depth (DOD) and precipitation remote-sensing estimates along the transect were obtained using Moderate
- 125 Resolution Imaging Spectroradiometer (MODIS) and Tropical Rainfall Measuring Mission (TRMM) datasets, respectively. These methods are described in detail in the Supporting Information.

3. Results and Discussion

130 **3.1 Mineral Dust Fluxes**

Residual-mass fluxes are often taken to represent the lithogenic or mineral dust fraction, and result from subtracting the weights of biogenic constituents from the total mass (e.g. Jickells et al., 1998; Fischer et al., 2016; Korte et al., 2017), defined as

where $CaCO_3$ represents calcium carbonates, $BSiO_2$ is biogenic silica, and OM is the organic matter. For the exact calculations, see Korte et al. (2017). All three fractions are determined by using specific calculation factors that are, due to, for example, an uncertain water content for $BSiO_2$ and OM,

- 140 globally not well constrained and the range is often not taken into account. Since this representation of the lithogenic or mineral dust fraction is obtained indirectly, this fraction is associated with several uncertainties related to the conversion factors and measurements of the individual biogenic fractions. Here, we present dust-mass fluxes which were obtained by directly weighing the dust fraction, and are therefore more accurate and representative for the dust flux. In our method, the biogenic components
- 145 were chemically removed and this resulted in the dust fraction only. These dust mass fluxes are significantly smaller than the residual-mass fluxes (Fig. 1) determined by Korte et al. (2017) on the same samples as were used here. Dust-mass fluxes at M1, M2 and M4 presented here are all highly correlated with the residual-mass flux (Korte et al., 2017), with R² values ranging between 0.69 and 0.78 and slopes decreasing downwind from 0.57 to 0.25 and 0.15, respectively. On average, the
- 150 residual-mass fluxes are 5.8 times higher than the dust mass fluxes (standard deviation = 3.3), corroborating the conclusion by Korte et al. (2017) that the residual fraction most likely overestimates dust fluxes due to various assumptions and uncertainties in calculating the masses of biogenic matter. Additional components of the residual fraction potentially include biogenic phosphates and sulfates, particles of volcanogenic, cosmogenic and anthropogenic origin, as well as crystal water associated
- 155 with the opaline silica and the OH content of organic matter and clay minerals (Korte et al., 2017). Although the residual fraction does reflect the temporal changes in dust deposition at each site, the absolute dust fluxes are severely overestimated and differ between sites. Alternative methods of estimating the dust deposition fluxes use particular lithogenic elements as proxies, mainly Al and Ti (e.g. Bory et al., 2002; Pabortsava et al., 2017), or automated mineral classification by EDX (Brust et
- 160 al., 2011), but these methods also rely on several assumptions involving the composition and size of the dust particles, and therefore direct measurements of dust deposition fluxes are considered to be more accurate.



165

Figure 1. Dust-mass flux (established here) versus residual-mass flux (data from Korte et al., 2017) for 2013-M1–M4. Black line represents 1:1 relation.

3.2 Dust deposition along the transect

Dust-mass fluxes are highest at M1, located about 700 km from the west African coast, reaching maximum values of >30 mg m⁻² day⁻¹ and an average value of 14.3 mg m⁻² day⁻¹ in 2013 (Fig. 2). This 170 is about 1.5 times higher than the lithogenic fluxes observed close to the Canary Islands by Neuer et al. (2004) and Brust et al. (2011), who observe average deposition fluxes of 10 and 9.5 mg m⁻² day⁻¹, and a maximum deposition flux in winter of up to 54 and 43 mg m² day⁻¹, respectively. The episodic high fluxes in winter are likely due to dust transport at lower altitudes during this time and therefore deposited closer to the source (Van der Does et al., 2016), while the overall lower fluxes probably 175 relate to the more northern position of the Canary Island traps, away from the center of the main dust cloud. In our sediment traps, dust deposition also decreases sharply towards the north, to 2.5 and 2.9 mg m⁻² day⁻¹ at M2 in 2013 and 2014, and towards the west to 3.7 mg m⁻² day⁻¹ at M3 in 2014, and 3.3 and 2.3 mg m⁻² day⁻¹ at M4 in 2013 and 2014 (Table 1). These fluxes are lower than the average 8 mg m^{-2} day⁻¹ deposition estimates by Pabortsava et al. (2017) in the oligotrophic gyre north of M2. This 180 sharp down-wind decrease was also observed in sediment traps offshore Mauritania by Bory et al. (2002), who found a decrease from 50 to 9 mg m⁻² day⁻¹, at 500 and 1500 km from the coast, respectively.

185 *Table 1. Average dust deposition fluxes in mg m⁻² day⁻¹ for all the sediment traps in 2013 and 2014. For 2014-M1 only the winter season was sampled completely.*

sediment trap	2013-M1	2013-M2	2013-M4	2014-M1	2014-M2	2014-M3	2014-M4
annual average	14.3	2.5	3.3		2.9	3.7	2.3
spring average	9.4	1.2	1.9		2.3	1.9	1.3
summer average	22.6	3.3	4.7		2.8	4.0	3.2
autumn average	15.4	3.0	4.5		4.2	5.8	3.8
winter average	8.5	2.4	1.8	9.7	2.5	3.3	1.5

- Satellite observations by CALIOP (Cloud-Aerosol Lidar with Orthogonal Polarization) show that an
 estimated 182 Tg of dust is emitted from the African continent at 15°W toward the Atlantic Ocean annually (Yu et al., 2015). At 35°W this atmospheric dust load is reduced to 132 Tg, and at 75°W to 43 Tg, which translates into losses of 14.0 and 12.4 mg m⁻² day⁻¹ between 15–35°W and 35–75°W, respectively. Although the satellite observations exhibit a great uncertainty of ±45-70% (Yu et al., 2015), their estimates are very similar to the average dust deposition observed for the eastern Atlantic in the sediment traps at M1. However, in the central and western Atlantic the observed deposition fluxes in the sediment traps M2–M4 are considerably lower than the fluxes derived from CALIOP
- estimates (12.4 mg m² day⁻¹). Instead of deposition into the ocean, this decrease in atmospheric dust load derived from CALIOP estimates may have resulted from dispersal of the dust cloud out of the studied area. Also, the recent publication by Yu et al. (2019) presents dust deposition derived from
- 200 different satellite products, which shows much higher deposition fluxes than observed in our sediment traps, for all four seasons, although the general seasonal and spatial trends are very similar. They also point out that the dust fluxes show high correlations between the different satellite products, but with large offsets in the magnitude of dust fluxes. They further state that dust deposition occurs further south than the observed DOD pattern due to the effective removal of dust by precipitation at the
- 205 southern edge of dust plumes, related to the ITCZ (Yu et al., 2019). This highlights the importance of *in situ* observations of dust deposition over the Atlantic, since most open-ocean records of dust transport and deposition are either indirect observations, for example by remote sensing, or simulated by models, which both require validation against *in situ* records.



210

Figure 2. Time series of Saharan dust and precipitation across the Atlantic. A-D: Mass fluxes of Saharan mineral dust (particles >0.4µm) collected by sediment traps (bars), DOD (orange lines), and precipitation (TRMM; blue lines), at stations M1–M4. Width of the gray bars corresponds to the respective 8- or 16-day collection intervals. Blue and red areas mark winter and summer seasons, respectively.

215

Seasonality of dust deposition observed in the sediment traps is well defined across the Atlantic, with highest fluxes in summer and autumn, and lowest fluxes in winter and spring (Table 1). The seasonal amplitude decreases downwind, from east to west, with largest seasonal differences observed at station M1, in line with the downwind decrease in seasonality observed in the particle size of the deposited

- 220 dust (Van der Does et al., 2016). Satellites have also observed highest dust fluxes in the Atlantic in summer, although contrary to the observations in the sediment traps, the satellites observed lowest fluxes in autumn and reaches less far west (Yu et al., 2019). Consistent with increased deposition fluxes in the traps, atmospheric dust loads measured as DOD were also highest in early summer (Fig. 2). As virtually no precipitation occurs along the transect from winter until June (Fig. 2), only dry
- 225 deposition can account for the fluxes intercepted during this time by the sediment traps. Later in summer, DOD decreases at the same time when precipitation increases (June-July), although deposition fluxes in the traps remain high. This suggests that the dust is washed out of the atmosphere through precipitation (Fig. 2), resulting in low DOD values, but high (wet-)deposition fluxes. This

trend continued until November, and this anti-correlation between atmospheric dust load and dust deposition in some regions and seasons is consistent with model simulations (Mahowald et al., 2003).

230 deposition in some regions and seasons is consistent with model simulations (Mahowald et al., 2003). In winter, wet deposition likely occurs more to the south than the studied transect, close to the ITCZ (Yu et al., 2019).

If the dust is indeed deposited by wet deposition, the satellite data suggest a complete washout of
atmospheric dust at the first incidence of precipitation, not directly related to the quantity or duration of precipitation (Fig. 2). Bory et al. (2002) also found wet deposition of dust to dominate in summer, on the eastern side of the Atlantic, and found that wet deposition mainly takes place in a few major events, during which the majority of the annual deposition is established. Such major wet-deposition events were also found by Neuer et al. (2004), with deposition fluxes of 160 mg m⁻² day⁻¹. In short,
from combining sediment trap and satellite data we infer low amounts of dry deposition in winter and spring, and a sharp increase at the beginning of summer. This is followed by high amounts of wet deposition from late June until November.

Station M2 is positioned about 160 km north of station M3 and received lower dust fluxes and less
seasonality (Table 1), although multiple-year records, especially at M3, should be considered for a better understanding of seasonal differences between these sites. Less seasonality at M2 could result from the larger distance from the center of the Saharan dust plume. The position of the central axis of the main dust cloud is closely linked to the position of the ITCZ, which migrates seasonally (Ben-Ami et al., 2012; Meng et al., 2017).

250

270

Reported seasonality of dust concentrations and deposition is different on either side of the Atlantic, peaking in winter in the east (Ratmeyer et al., 1999; Bory et al., 2002; Neuer et al., 2004; Skonieczny et al., 2013; Fomba et al., 2014; Fischer et al., 2016), and in summer in the west (Jickells et al., 1998; Prospero et al., 2014). These differences can be related to seasonal atmospheric dynamics like the altitude at which the dust is transported, with dust at lower altitudes deposited close to the source in 255 the east, while dust transported at higher altitudes can reach the remote western Atlantic (Van der Does et al., 2016; Yu et al., 2019). In addition, most of the studies on the eastern side of the Atlantic were carried out at latitudes above the main dust cloud and the northern position of the ITCZ, which could result in the opposing seasonal signal compared to the western Atlantic. The sampling method 260 can also influence the seasonal signal of dust fluxes, as both dry and wet deposition are collected by the sediment traps, while only "dry" dust collected directly from the atmosphere is sampled at Barbados. The difference in dust deposition between sampling sites highlights the importance of dust collection between locations close to the source (eastern Atlantic) and further downwind (western Atlantic). The transect of sampled dust presented here shows year-round dust deposition across the 265 Atlantic, peaking in summer most prominently at the eastern end of the transect.

3.3 Modeled dust deposition

When comparing the model results to the observed deposition fluxes in the sediment traps, it shows that the model performs well in terms of simulating the seasonality of dust deposition, but performs less well in terms of simulating the magnitude of dust deposition along this transect. As Albani et al.

- (2014) report, the model overestimates dust deposition and dust surface concentrations in the North Atlantic, but the overall seasonality of dust concentration is captured well. Given the overall uncertainties in modeling the mineral dust cycle with earth system models (Huneeus et al., 2011), the model can still be considered a useful tool to put the observations used in our study into perspective.
- 275 The fluxes of total dust deposition (sum of wet and dry deposition, Fig. 3A) simulated by the earth system model show a similarly strong decrease from east to west, albeit more gradual than the

downwind decrease in fluxes observed in the sediment traps (Fig. 3B). More importantly, the model results show a sharp seasonal contrast between dry and wet deposition. Modeled wet deposition (blue bars) is highest in summer and autumn across the tropical Atlantic, related to precipitation activity

- 280 along with the meridional ITCZ movement (Nicholson, 2000). Modeled dry deposition (red bars) is highest in winter and spring in the east and decreases westward in both magnitude and seasonal amplitude. These simulated deposition maxima in winter and spring (dry deposition) and late summer and autumn (wet deposition) in the Atlantic Ocean are corroborated by the sediment-trap observations and the incidence of precipitation, with the (wet) deposition maxima in summer being the highest (Fig.
- 3B). From east to west, the relative contribution of dry deposition to total deposition in winter and spring decreases, making wet deposition in summer and autumn the dominant modeled deposition mode in particular in the western Atlantic. It accounts for most of the total annual dust deposition there and determines the seasonal signature of total deposition. This result also matches dust deposition observed in Florida at the western side of the Atlantic, where wet deposition was found to dominate
 (Prospero et al., 2010).



Figure 3. A: Simulated trans-Atlantic dust deposition fluxes (multi-year 5°×10° lat-lon averages in dashed boxes on map): dry deposition (red bars) stacked upon wet deposition (blue bars), their sum displays the total dust deposition (dry + wet). B: Annual cycle of dust deposition in the submarine sediment traps M1–M4 (bars; right Y-axes) with monthly precipitation (TRMM; blue lines; left Y-axes), for 2013–2014. C: Annual cycle of atmospheric dust concentrations at Barbados (bars; right Y-axis) and averaged monthly precipitation (TRMM, blue line; left Y-axis) for 2013–2014. The map was created using Ocean Data View (Schlitzer, 2018).

3.4 Dust concentrations at Barbados

300

On Barbados in the far western Atlantic, atmospheric dust loads are highest in summer and autumn (Fig. 3C), which is in line with the long-term observational record on Barbados (Prospero et al., 2014). Given that wet deposition dominates over the Atlantic during summer and autumn, the Saharan dust

- 305 arriving at Barbados is therefore likely the remaining dust not washed out by rains during its trans-Atlantic crossing, and to a smaller extent dry deposition. This is also evidenced by the atmospheric dust concentrations at Barbados that peak (June) before precipitation peaks across the Atlantic (August–October; Fig. 3). Prospero and Nees (1986) have initially found a relationship between Barbados dust concentrations and previous-year Sahel precipitation, although additional data of later
- 310 studies did not support this relationship anymore (e.g. Prospero, 1996). Our results demonstrate that dust concentrations at Barbados could be affected not only by precipitation-related processes in the source region, but also by precipitation during long-range transport across the Atlantic Ocean, independent of dust generation and emission. Given our current findings, we argue that precipitation over the North Atlantic could be another important factor in influencing Saharan dust *transport* across
- 315 the Atlantic and *concentrations* over Central America and the Caribbean. Investigating the relation between Barbados dust concentrations and rainfall over the Atlantic over a longer time scale should shed more light on the dominance of wet deposition over the Atlantic Ocean, and how this influences dust concentrations in the Caribbean.

320 **3.5 Ocean fertilization**

The results presented here show that most Saharan dust over the tropical North Atlantic Ocean is deposited with precipitation. Wet-dust deposition has important implications for transferring atmospheric dust-borne nutrients into the ocean, as low pH conditions in aerosols and clouds favor the leaching of nutrients from the dust particles (Desboeufs et al., 2001; Pulido-Villena et al., 2014; Korte

- 325 et al., 2018). This may stimulate phytoplankton production, which in turn impacts atmospheric CO₂ levels (Mills et al., 2004; Guieu et al., 2014; Jickells et al., 2014; Pabortsava et al., 2017). Bottle-incubation experiments with Saharan dust additions along the same transect as in the present study have shown that Saharan dust indeed has the potential to deliver macro- and micronutrients (e.g. phosphate, silica, and dissolved iron) if exposed to acidic conditions before deposition (Korte et al., 2018).
- 330 2018). Not surprisingly, the concentration of nutrients released is in turn related to the amount of dust deposited, highlighting the importance for accurate dust flux measurements.

4. Conclusions

Our combined results of *in situ* observations of dust deposition fluxes, model simulations and remote sensing provide deeper insights into Saharan dust deposition across the Atlantic Ocean. Direct observations on dust deposition in such remote areas are sparse but essential for model calibrations, which underscores the uniqueness of the current dataset. We show that dust transport and deposition are highest during summer and demonstrate that wet deposition is the main mechanism for depositing Saharan dust into the tropical Atlantic Ocean: nearly all dust deposition in the western Atlantic

- 340 originates from wet deposition during summer. As a result, dust arriving in the Caribbean is the remainder of dust escaping wash-out by precipitation along its passage from east to west. Wetdeposition events may lead to increased release of macro- and micronutrients that are important for the ocean's carbon cycle. Consequently, wet deposition of bio-available nutrients from mineral dust may be driving increased surface-ocean productivity, particularly in oligotrophic parts of the world's
- 345 oceans. Considering future changes in global and regional precipitation patterns, enhanced droughts and the many uncertainties regarding the dust cycle, an improved understanding of these mechanisms is key to future climate predictions and to a better understanding of the impact of mineral dust on global carbon budgets through ocean fertilization. In addition, our findings could be used for the interpretation of sediment archives and long-term paleo dust records in the ocean.

Acknowledgments

This project was funded by NWO (project 822.01.008, TRAFFIC), and ERC (project 311152, DUSTTRAFFIC) awarded to JBS. The work of UM has been performed within the framework of the

- 355 PalMod project (FKZ: 01LP1508B), supported by the German Federal Ministry of Education and Research (BMBF) as Research for Sustainability initiative (FONA). The CESM experiment was performed with resources provided by the North-German Supercomputing Alliance (HLRN). HY was supported by the NASA CALIPSO/CloudSat project managed by David Considine. The captains, crews, and scientists of FS *Meteor* cruise M89, RV *Pelagia* cruises 64PE378 and 64PE395, and NIOZ
- 360 technicians are thanked for deployment and retrieval of the sediment-trap moorings. Chris Munday, Piet van Gaever and Juliane Steinhardt are thanked for their assistance in sediment-trap sample processing. The authors thank Joseph M. Prospero for providing the samples collected at Barbados. Claudia Alvarez is thanked for assistance with obtaining the Barbados dust concentrations, and processing of the filters was supported by PZ using University of Miami faculty funds. TRMM
- 365 precipitation data were obtained from the Giovanni online data system, developed and maintained by the NASA Goddard Earth Sciences Data and Information Services Center (GES DISC).

Data Availability

All data used in this manuscript are stored at https://doi.org/10.1594/PANGAEA.912241

References

375	Adams, A. M., Prospero, J. M., & Zhang, C. (2012). CALIPSO-Derived Three-Dimensional Structure of Aerosol over the Atlantic Basin and Adjacent Continents. <i>Journal of Climate, 25</i> (19), 6862-6879. doi: <u>https://doi.org/10.1175/JCLI-D-11-00672</u> 1
0,0	Albani, S., Mahowald, N. M., Perry, A. T., Scanza, R. A., Zender, C. S., Heavens, N. G., Maggi, V., Kok, J. F., & Otto- Bliesner, B. L. (2014). Improved dust representation in the Community Atmosphere Model. <i>Journal of Advances in</i> <i>Modeling Earth Systems</i> , 6(3), 541-570. doi: <u>https://doi.org/10.1002/2013MS000279</u>
380	Atkinson, J. D., Murray, B. J., Woodhouse, M. T., Whale, T. F., Baustian, K. J., Carslaw, K. S., Dobbie, S., O/Sullivan, D., & Malkin, T. L. (2013). The importance of feldspar for ice nucleation by mineral dust in mixed-phase clouds. <i>Nature</i> , 498(7454), 355-358. doi: <u>https://doi.org/10.1038/nature12278</u>
205	Baker, A. R., Lesworth, T., Adams, C., Jickells, T. D., & Ganzeveld, L. (2010). Estimation of atmospheric nutrient inputs to the Atlantic Ocean from 50°N to 50°S based on large-scale field sampling: Fixed nitrogen and dry deposition of phosphorus. <i>Global Biogeochemical Cycles</i> , 24(3). doi: <u>https://doi.org/10.1029/2009GB003634</u>
385	Ben-Ami, Y., Koren, I., Altaratz, O., Kostinski, A., & Lehahn, Y. (2012). Discernible rhythm in the spatio/temporal distributions of transatlantic dust. <i>Atmos. Chem. Phys.</i> , <i>12</i> (5), 2253-2262. doi: <u>https://doi.org/10.5194/acp-12-2253-2012</u>
	Bergametti, G., & Forêt, G. (2014). Dust Deposition. In P. Knippertz & J. B. W. stuut (Eds.), <i>Mineral Dust, A Key Player in the Earth System</i> (pp. 179-200). Springer Science+Business Media Dordrecht: Springer.
390	Bory, A., Dulac, F., Moulin, C., Chiapello, I., Newton, P. P., Guelle, W., Lambert, C. E., & Bergametti, G. (2002). Atmospheric and oceanic dust fluxes in the northeastern tropical Atlantic Ocean: how close a coupling? <i>Annales Geophysicae</i> , 20(12), 2067-2076. doi:https://doi.org/10.5194/angeo-20-2067-2002
395	Bory, A., & Newton, P. P. (2000). Transport of airborne lithogenic material down through the water column in two contrasting regions of the eastern subtropical North Atlantic Ocean. <i>Global Biogeochemical Cycles</i> , 14(1), 297- 315. doi:https://doi.org/10.1029/1999GB900098
	Bressac, M., Guieu, C., Doxaran, D., Bourrin, F., Desboeufs, K., Leblond, N., & Ridame, C. (2014). Quantification of the Lithogenic Carbon Pump Following a Simulated Dust-Deposition Event in Large Mesocosms. <i>Biogeosciences</i> , 11(4), 1007-1020. doi:https://doi.org/10.5194/be-11-1007-2014
400	Brust, J., Schulz-Bull, D. E., Leipe, T., Chavagnac, V., & Waniek, J. J. (2011). Descending particles: From the atmosphere to the deep ocean—A time series study in the subtropical NE Atlantic. <i>Geophysical Research Letters</i> , 38(6). doi:https://doi.org/10.1029/2010GL045399
	Carlson, T. N., & Prospero, J. M. (1972). The Large-Scale Movement of Saharan Air Outbreaks over the Northern Equatorial Atlantic. <i>Journal of Applied Meteorology</i> , <i>11</i> (2), 283-297. doi: <u>https://doi.org/10.1175/1520-0450(1972)011<0283:TL SMOS>2 0 CO:2</u>
405	Desboeufs, K. V., Losno, R., & Colin, J. L. (2001). Factors influencing aerosol solubility during cloud processes. <i>Atmospheric Environment</i> , 35(20), 3529-3537. doi: <u>http://dx.doi.org/10.1016/S1352-2310(00)00472-6</u>
	Fischer, G., Romero, O., Merkel, U., Donner, B., Iversen, M., Nowald, N., Ratmeyer, V., Ruhland, G., Klann, M., & Wefer, G. (2016). Deep ocean mass fluxes in the coastal upwelling off Mauritania from 1988 to 2012: variability on
410	 Seasonal to decadal timescales. <i>Biogeosciences</i>, 13(10), 30/1-3090. doi:<u>https://doi.org/10.5194/bg-13-30/1-2016</u> Fomba, K. W., Müller, K., van Pinxteren, D., Poulain, L., van Pinxteren, M., & Herrmann, H. (2014). Long-term chemical characterization of tropical and marine aerosols at the Cape Verde Atmospheric Observatory (CVAO) from 2007 to 2011. <i>Atmos. Chem. Phys.</i>, 14(17), 8883-8904. doi:10.5194/acp-14-8883-2014
	Goudie, A. S., & Middleton, N. J. (2001). Saharan Dust Storms: Nature and Consequences. <i>Earth-Science Reviews</i> , 56(1-4), 179-204. doi:https://doi.org/10.1016/S0012-8252(01)00067-8
415	Guieu, C., Aumont, O., Paytan, A., Bopp, L., Law, C. S., Mahowald, N., Achterberg, E. P., Marañón, E., Salihoglu, B., Crise, A., Wagener, T., Herut, B., Desboeufs, K., Kanakidou, M., Olgun, N., Peters, F., Pulido-Villena, E., Tovar- Sanchez, A., & Völker, C. (2014). The significance of the episodic nature of atmospheric deposition to Low
	Nutrient Low Chlorophyll regions. <i>Global Biogeochemical Cycles</i> , 28(11), 1179-1198. doi:doi:10.1002/2014GB004852
420	Huneeus, N., Schulz, M., Balkanski, Y., Griesfeller, J., Prospero, J., Kinne, S., Bauer, S., Boucher, O., Chin, M., Dentener, F., Diehl, T., Easter, R., Fillmore, D., Ghan, S., Ginoux, P., Grini, A., Horowitz, L., Koch, D., Krol, M. C., Landing, W., Liu, X., Mahowald, N., Miller, R., Morcrette, J. J., Myhre, G., Penner, J., Perlwitz, J., Stier, P., Takemura, T., & Zender, C. S. (2011). Global dust model intercomparison in AeroCom phase I. <i>Atmospheric</i> <i>Chemistry and Physics</i> , 11(15), 7781-7816, doi:https://doi.org/10.5194/acp-11-7781-2011
425	Hurrell, J. W., Holland, M. M., Gent, P. R., Ghan, S., Kay, J. E., Kushner, P. J., Lamarque, JF., Large, W. G., Lawrence, D., Lindsay, K., Lipscomb, W. H., Long, M. C., Mahowald, N., Marsh, D. R., Neale, R. B., Rasch, P., Vavrus, S., Vertenstein, M., Bader, D., Collins, W. D., Hack, J. J., Kiehl, J., & Marshall, S. (2013). The Community Earth System Model: A Framework for Collaborative Research. <i>Bulletin of the American Meteorological Society</i> , 94(9), 1330–1360. doi:https://doi.org/10.1175/PAMS.D.12.00121.1
430	Jickells, T. D., Boyd, P. W., & Hunter, K. A. (2014). Biogeochemical Impacts of Dust on the Global Carbon Cycle. In P. Knippertz & J. B. W. Stuut (Eds.), <i>Mineral Dust, A Key Player in the Earth System</i> (pp. 359-384). Springer Science+Business Media Dordrecht
	Jickells, T. D., Dorling, S., Deuser, W. G., Church, T. M., Arimoto, R., & Prospero, J. M. (1998). Air-borne dust fluxes to a deep water sediment trap in the Sargasso Sea <i>Global Biogeochemical Cycles</i> 12(2) 311-320
435	doi:10.1029/97gb03368

- Johnson, M. S., & Meskhidze, N. (2013). Atmospheric dissolved iron deposition to the global oceans: effects of oxalatepromoted Fe dissolution, photochemical redox cycling, and dust mineralogy. *Geosci. Model Dev.*, 6(4), 1137-1155. doi:10.5194/gmd-6-1137-2013
- Korte, L. F., Brummer, G. J. A., van der Does, M., Guerreiro, C. V., Hennekam, R., van Hateren, J. A., Jong, D., Munday, C. I., Schouten, S., & Stuut, J. B. W. (2017). Downward particle fluxes of biogenic matter and Saharan dust across the equatorial North Atlantic. *Atmos. Chem. Phys.*, 17(9), 6023-6040. doi:https://doi.org/10.5194/acp-17-6023-2017
 - Korte, L. F., Pausch, F., Trimborn, S., Brussaard, C. P. D., Brummer, G. J. A., van der Does, M., Guerreiro, C. V., Schreuder, L. T., Munday, C. I., & Stuut, J. B. W. (2018). Effects of dry and wet Saharan dust deposition in the tropical North Atlantic Ocean. *Biogeosciences Discuss.*, 2018, 1-20. doi:<u>https://doi.org/10.5194/bg-2018-484</u>
- 445 Mahowald, N., Albani, S., Kok, J. F., Engelstaeder, S., Scanza, R., Ward, D. S., & Flanner, M. G. (2014). The Size Distribution of Desert Dust Aerosols and its Impact on the Earth System. *Aeolian Research*, 15, 53-71. doi:<u>https://doi.org/10.1016/j.aeolia.2013.09.002</u>
 - Mahowald, N., Luo, C., del Corral, J., & Zender, C. S. (2003). Interannual variability in atmospheric mineral aerosols from a 22-year model simulation and observational data. *Journal of Geophysical Research: Atmospheres, 108*(D12). doi:10.1029/2002jd002821
 - Meng, L., Gao, H. W., Yu, Yu, Yao, X. H., Gao, Y., Zhang, C., & Fan, L. (2017). A new approach developed to study variability in North African dust transport routes over the Atlantic during 2001–2015. *Geophysical Research Letters*, 44(19), 10,026-010,035. doi:10.1002/2017GL074478
- Meskhidze, N., Chameides, W. L., & Nenes, A. (2005). Dust and pollution: A recipe for enhanced ocean fertilization?
 Journal of Geophysical Research: Atmospheres, 110(D3). doi:10.1029/2004jd005082
 - Mills, M. M., Ridame, C., Davey, M., La Roche, J., & Geider, R. J. (2004). Iron and phosphorus co-limit nitrogen fixation in the eastern tropical North Atlantic. *Nature*, 429, 292. doi:<u>https://doi.org/10.1038/nature02550</u>
 - Muhs, D. R. (2013). The geologic records of dust in the Quaternary. *Aeolian Research*, *9*, 3-48. doi:<u>https://doi.org/10.1016/j.aeolia.2012.08.001</u>
- 460 Neuer, S., Torres-Padrón, M. E., Gelado-Caballero, M. D., Rueda, M. J., Hernández-Brito, J., Davenport, R., & Wefer, G. (2004). Dust deposition pulses to the eastern subtropical North Atlantic gyre: Does ocean's biogeochemistry respond? *Global Biogeochemical Cycles*, 18(4). doi:<u>https://doi.org/10.1029/2004GB002228</u>
 - Nicholson, S. E. (2000). The Nature of Rainfall Variability over Africa on Time Scales of Decades to Millenia. *Global and Planetary Change*, 26(1-3), 137-158. doi:<u>https://doi.org/10.1016/S0921-8181(00)00040-0</u>
- 465 Pabortsava, K., Lampitt, R. S., Benson, J., Crowe, C., McLachlan, R., Le Moigne, F. A. C., Mark Moore, C., Pebody, C., Provost, P., Rees, A. P., Tilstone, Gavin H., & Woodward, E. M. S. (2017). Carbon sequestration in the deep Atlantic enhanced by Saharan dust. *Nature Geoscience*, 10, 189. doi:<u>https://doi.org/10.1038/ngeo2899</u>
 - Prospero, J. M. (1996). Saharan Dust Transport over the North Atlantic Ocean and Mediterranean: an Overview. In S. Guerzoni & R. Chester (Eds.), *The impact of desert dust across the Mediterranean* (Vol. 11, pp. 133-151). Dordrecht / Boston / London: Kluwer Academic.
 - Prospero, J. M., Collard, F.-X., Molinié, J., & Jeannot, A. (2014). Characterizing the annual cycle of African dust transport to the Caribbean Basin and South America and its impact on the environment and air quality. *Global Biogeochemical Cycles*, 28(7), 757-773. doi:<u>https://doi.org/10.1002/2013GB004802</u>
- Prospero, J. M., Landing, W. M., & Schulz, M. (2010). African dust deposition to Florida: Temporal and spatial variability and comparisons to models. *Journal of Geophysical Research: Atmospheres, 115*(D13). doi:<u>https://doi.org/10.1029/2009JD012773</u>
 - Prospero, J. M., & Nees, R. T. (1986). Impact of the North African Drought and El-Nino on Mineral Dust in the Barbados Trade Winds. *Nature*, 320(6064), 735-738. doi:<u>https://doi.org/10.1038/320735a0</u>
- Pulido-Villena, E., Baudoux, A. C., Obernosterer, I., Landa, M., Caparros, J., Catala, P., Georges, C., Harmand, J., & Guieu,
 C. (2014). Microbial food web dynamics in response to a Saharan dust event: results from a mesocosm study in the oligotrophic Mediterranean Sea. *Biogeosciences*, *11*(19), 5607-5619. doi:<u>https://doi.org/10.5194/bg-11-5607-2014</u>
 - Ratmeyer, V., Fischer, G., & Wefer, G. (1999). Lithogenic Particle Fluxes and Grain Size Distributions in the Deep Ocean off Northwest Africa: Implications for Seasonal Changes of Aeolian Dust Input and Downward Transport. Deep-Sea Research Part I-Oceanographic Research Papers, 46(8), 1289-1337. doi:<u>https://doi.org/10.1016/S0967-0637(99)00008-4</u>
 - Ridame, C., Dekaezemacker, J., Guieu, C., Bonnet, S., L'Helguen, S., & Malien, F. (2014). Contrasted Saharan dust events in LNLC environments: impact on nutrient dynamics and primary production. *Biogeosciences*, 11(17), 4783-4800. doi:<u>https://doi.org/10.5194/bg-11-4783-2014</u>
- Ryder, C. L., Highwood, E. J., Rosenberg, P. D., Trembath, J., Brooke, J. K., Bart, M., Dean, A., Crosier, J., Dorsey, J., Brindley, H., Banks, J., Marsham, J. H., McQuaid, J. B., Sodemann, H., & Washington, R. (2013). Optical properties of Saharan dust aerosol and contribution from the coarse mode as measured during the Fennec 2011 aircraft campaign. *Atmospheric Chemistry and Physics*, 13(1), 303-325. doi:<u>https://doi.org/10.5194/acp-13-303-</u>2013
- 495 Sarnthein, M., Tetzlaff, G., Koopmann, B., Wolter, K., & Pflaumann, U. (1981). Glacial and Interglacial Wind Regimes over the Eastern Subtropical Atlantic and North-West Africa. *Nature*, 293(5829), 193-196. doi:<u>https://doi.org/10.1038/293193a0</u>
 - Schlitzer, R. (2018). Ocean Data View. See https://odv.awi.de.
- Shao, Y. P., Wyrwoll, K. H., Chappell, A., Huang, J. P., Lin, Z. H., McTainsh, G. H., Mikami, M., Tanaka, T. Y., Wang, X. L., & Yoon, S. (2011). Dust Cycle: An Emerging Core Theme in Earth System Science. *Aeolian Research*, 2(4), 181-204. doi:<u>https://doi.org/10.1016/j.aeolia.2011.02.001</u>
- Skonieczny, C., Bory, A., Bout-Roumazeilles, V., Abouchami, W., Galer, S. J. G., Crosta, X., Diallo, A., & Ndiaye, T. (2013). A Three-Year Time Series of Mineral Dust Deposits on the West African Margin: Sedimentological and

440

450

470

Geochemical Signatures and Implications for Interpretation of Marine Paleo-Dust Records. *Earth and Planetary Science Letters*, 364(0), 145-156. doi:<u>https://doi.org/10.1016/j.epsl.2012.12.039</u>

- 505 Spokes, L. J., Jickells, T. D., & Lim, B. (1994). Solubilisation of aerosol trace metals by cloud processing: A laboratory study. *Geochimica Et Cosmochimica Acta*, 58(15), 3281-3287. doi:<u>https://doi.org/10.1016/0016-7037(94)90056-6</u>
 - Stuut, J. B., Zabel, M., Ratmeyer, V., Helmke, P., Schefuss, E., Lavik, G., & Schneider, R. (2005). Provenance of Present-Day Eolian Dust Collected off NW Africa. *Journal of Geophysical Research-Atmospheres*, 110(D04202). doi:<u>https://doi.org/10.1029/2004JD005161</u>
- 510 Tsamalis, C., Chedin, A., Pelon, J., & Capelle, V. (2013). The seasonal vertical distribution of the Saharan Air Layer and its modulation by the wind. *Atmospheric Chemistry and Physics*, 13(22), 11235-11257. doi:https://doi.org/10.5194/acp-13-11235-2013
 - Twohy, C. H., Kreidenweis, S. M., Eidhammer, T., Browell, E. V., Heymsfield, A. J., Bansemer, A. R., Anderson, B. E., Chen, G., Ismail, S., DeMott, P. J., & Van Den Heever, S. C. (2009). Saharan dust particles nucleate droplets in eastern Atlantic clouds. *Geophysical Research Letters*, 36(1). doi:<u>https://doi.org/10.1029/2008GL035846</u>
- 515 eastern Atlantic clouds. *Geophysical Research Letters*, 36(1). doi:<u>https://doi.org/10.1029/2008GL035846</u>
 Van der Does, M., Knippertz, P., Zschenderlein, P., Giles Harrison, R., & Stuut, J.-B. W. (2018). The mysterious long-range transport of giant mineral dust particles. *Science Advances*, 4(12), eaau2768. doi:<u>https://doi.org/10.1126/sciadv.aau2768</u>
- Van der Does, M., Korte, L. F., Munday, C. I., Brummer, G. J. A., & Stuut, J. B. W. (2016). Particle size traces modern Saharan dust transport and deposition across the equatorial North Atlantic. *Atmos. Chem. Phys.*, 16(21), 13697-13710. doi:<u>https://doi.org/10.5194/acp-16-13697-2016</u>
 - Van der Jagt, H., Friese, C., Stuut, J. B., Fischer, G., & Iversen, M. (2018). The ballasting effect of Saharan dust deposition on aggregate dynamics and carbon export: aggregation, settling and scavenging of marine snow. *Limnology and Oceanography, 63*, 1386-1394. doi:<u>https://doi.org/10.1002/lno.10779</u>
- Weinzierl, B., Ansmann, A., Prospero, J. M., Althausen, D., Benker, N., Chouza, F., Dollner, M., Farrell, D., Fomba, W. K., Freudenthaler, V., Gasteiger, J., Groß, S., Haarig, M., Heinold, B., Kandler, K., Kristensen, T. B., Mayol-Bracero, O. L., Müller, T., Reitebuch, O., Sauer, D., Schäfler, A., Schepanski, K., Spanu, A., Tegen, I., Toledano, C., & Walser, A. (2017). The Saharan Aerosol Long-Range Transport and Aerosol–Cloud-Interaction Experiment: Overview and Selected Highlights. *Bulletin of the American Meteorological Society*, *98*(7), 1427-1451.
 doi:https://doi.org/10.1175/BAMS-D-15-00142.1
- Yu, H., Chin, M., Bian, H., Yuan, T., Prospero, J. M., Omar, A. H., Remer, L. A., Winker, D. M., Yang, Y., Zhang, Y., & Zhang, Z. (2015). Quantification of Trans-Atlantic Dust Transport from Seven-Year (2007–2013) Record of CALIPSO Lidar Measurements. *Remote Sensing of Environment*, 159(0), 232-249. doi:https://doi.org/10.1016/j.rse.2014.12.010
- 535 Yu, H., Tan, Q., Chin, M., Remer, L. A., Kahn, R. A., Bian, H., Kim, D., Zhang, Z., Yuan, T., Omar, A. H., Winker, D. M., Levy, R. C., Kalashnikova, O., Crepeau, L., Capelle, V., & Chédin, A. (2019). Estimates of African Dust Deposition Along the Trans-Atlantic Transit Using the Decadelong Record of Aerosol Measurements from CALIOP, MODIS, MISR, and IASI. *Journal of Geophysical Research: Atmospheres, 124*(14), 7975-7996. doi:10.1029/2019jd030574
- 540 Zender, C. S., Bian, H., & Newman, D. (2003). Mineral Dust Entrainment and Deposition (DEAD) model: Description and 1990s dust climatology. *Journal of Geophysical Research: Atmospheres, 108*(D14). doi:<u>https://doi.org/10.1029/2002JD002775</u>